Author’s response to reviews

Title: Family physician views about primary care reform in Ontario: a postal questionnaire

Authors:

Dr Duncan J W Hunter (hunter@post.queensu.ca)
Samuel E D Shortt (seds@post.queensu.ca)
Peter M Walkert (petewalker007@yahoo.ca)
Dr Marshall Godwin (godwinm@post.queensu.ca)

Version: 4 Date: 25 Feb 2004

Formatting changes:

1) Country affiliations for each author added.

2) Changed statement on title pages to "All authors contributed equally to this work"

3) Blank page 17 removed

4) Published references provided in journal style. Left 'Grey literature' references as is.